Author's response to reviews

Title: A population-based study of race-specific risk for placental abruption

Authors:

Tammy T Shen (shent@msnotes.wustl.edu)
Emily A DeFranco (defrancoe@wudosis.wustl.edu)
David M Stamilio (stamiliod@wudosis.wustl.edu)
Jen Jen Chang (jjchang@slu.edu)
Louis J Muglia (muglia_l@kids.wustl.edu)

Version: 2 Date: 24 May 2008

Author's response to reviews:

Dear Dr. Bucceri,

Thank you for the opportunity to revise the abstract for our manuscript entitled,"A population-based study of race-specific risk for placental abruption" under consideration at BMC Pregnancy and Childbirth.

We have added more description of the background and aims of the study as requested, and also have provided greater detail of the methods, results and conclusions in accordance with the abstract guidelines. We hope you find this additional information useful.

Sincerely,

Louis Muglia, MD PhD